Author's response to reviews

Title: Public awareness of cancer risk factors and its determinants in Morocco: A population-based cross-sectional study

Authors:

Karima El Rhazi (elrhazikarima@gmail.com)
Bahia Bennani (bahia_bc@yahoo.fr)
Samira Elfakir (elfakirsamira@yahoo.fr)
Ahmadou Boly (ahmadouboly@hotmail.fr)
Ahmed Zidouh (azidouhma@yahoo.fr)
Rachid Bekkali (bekko26@yahoo.fr)
Chakib Nejjari (onejjari2000@yahoo.fr)

Version: 2 Date: 31 January 2013

Author's response to reviews: see over
COVER LETTER FOR SUBMISSION OF MANUSCRIPT

Karima EL RHAZI, MD, PhD
Epidemiologist
Department of Epidemiology and Public Health
Faculty of Medicine of Fez
Fez 30000
Morocco

Subject: SUBMISSION OF A MANUSCRIPT FOR EVALUATION

Dear Editor,

I am enclosing here with a manuscript entitled “Public awareness of risk factors for cancer and its determinants in Morocco: a population-based cross-sectional study.” for publication as a Research Article in “BMC Cancer” for evaluation.

Thank you for sending these comments. We have modified the manuscript according to the specific comments. Please find below our point by point response to the specific comments:

(1) Ethics Committee:
For approval ethics committee, this statement was added in the Methods section of the manuscript, including the name of the body which gave approval:

It was approved by the ethic committee of Fez university hospital center. All subjects gave their consent before answering the survey.

(2) Tables:
Tables were removed from the submission system and included within the text file of the manuscript after the references.
Table titles were also removed within the text.

Fes, January 31, 2013

Karima EL RHAZI